Author's response to reviews

Title: Severe bronchopulmonary dysplasia improved by noninvasive positive pressure ventilation: a case report

Authors:

Christian Mann (christian.mann@ksgr.ch)
Walter Bär (walter.baer@ksgr.ch)

Version: 7 Date: 21 June 2011

Author's response to reviews:

The following amendments were added:
1. Abstract on page 2 of manuscript: added
2. Sub-headings from case-presentation: removed

! The manuscript was revised by an English language translation and writing service before submission on 20 June 2011 !